Author's response to reviews

Title: The prevalence and correlates of alcohol use and alcohol use disorders: a population based study in Colombo, Sri Lanka

Authors:

Helena M.S. Zavos (helena.zavos@kcl.ac.uk)
Sisira Siribaddana (nipuna@stmail.lk)
Harriet A Ball (harriet.ball@gmail.com)
Micheal T Lynskey (micheal.lynskey@kcl.ac.uk)
Athula Sumathipala (athula.sumathipala@kcl.ac.uk)
Fruhling V Rijsdijk (fruhling.rijsdijk@kcl.ac.uk)
Matthew Hotopf (matthew.hotopf@kcl.ac.uk)

Version: 6 Date: 30 June 2015

Author's response to reviews:

We are delighted that you have provisionally accepted this manuscript for publication.

We have addressed the following issues and have attached a revised manuscript:

Please add all co-authors email addresses in the title page of your manuscript.

2. Competing Interest

Please remove the competing interest statement from abstract and title page of your manuscript.

3. Tables
Please ensure that the tables are cited in the correct order in the text.

Kind Regards,
Helena Zavos